

Check for updates

Effect of UK Quality and Outcomes Framework pay-for-performance programme on quality of primary care: systematic review with quantitative synthesis

Leonard Ho,¹ Stewart W Mercer,² David Henderson,³ Eddie Donaghy,³ Bruce Guthrie²

¹NIHR Health Determinants Research Collaboration

Aberdeen, Aberdeen, UK ²Advanced Care Research Centre, Usher Institute, University of Edinburgh, Edinburgh, UK ³Centre for Population Health Sciences, Usher Institute, University of Edinburgh, Edinburgh, UK

Correspondence to: B Guthrie bruce.guthrie@ed.ac.uk; (ORCID 0000-0003-4191-4880) Additional material is published online only. To view please visit

the journal online. Cite this as: BMJ 2025;389:e083424

http://dx.doi.org/10.1136/ bmj-2024-083424

Accepted: 27 May 2025

ABSTRACT

OBIECTIVE

To systematically review the one and three year impact on quality of care of the introduction and withdrawal of financial incentives in the UK Quality and Outcomes Framework pay-for-performance programme.

DESIGN

Systematic review with quantitative synthesis.

DATA SOURCES

MEDLINE, Embase, CINAHL, PsycINFO, and Scopus databases were searched from 1 January 2004 to 3 September 2024.

STUDY SELECTION

Eligible studies used repeated cross sectional or cohort designs with consistent measurement before and after incentive introduction or withdrawal. Studies with a minimum of three time points before and after intervention were included in quantitative synthesis.

DATA EXTRACTION

Analysis used both reported impact if available and de novo interrupted time series analysis of extracted raw data if not reported by the original study.

DATA SYNTHESIS

Meta-analysis was not appropriate; findings were quantitatively synthesised by reporting medians and interquartile ranges of changes in quality or reported narratively. Risk of bias was assessed using the Mixed Methods Assessment Tool.

RESULTS

30 studies were included, with 11 providing data for quantitative synthesis. Across all indicators, evidence was found of improvement in recorded quality at one year after incentive introduction (83 indicators;

WHAT IS ALREADY KNOWN ON THIS TOPIC

The UK Quality and Outcomes Framework (QOF) is one of the largest and longest running healthcare pay-for-performance programmes in the world

Pay for performance in healthcare has been deployed in many countries, but its effectiveness is still uncertain

Differences in context make comparing pay-for-performance programmes difficult

WHAT THIS STUDY ADDS

Introduction of QOF incentives was associated with an immediate improvement in recorded quality at one year compared with previous trends but not by three years

Withdrawal of incentives was associated with reductions in recorded quality at both one and three years, which were similar in size to the gains observed with incentive introduction

Whether or how best to deploy financial incentives in primary care therefore remains uncertain

at three years (72 indicators; median change 0.7 (-2.1 to 8.9) percentage points). Impact was higher for process indicators with lower performance in the

year before incentivisation. Incentive withdrawal was associated with reduction in recorded guality compared with predicted at both one year (31 indicators; median change -10.7 (IQR -17.9 to -3.8) percentage points) and three years (31 indicators; median change -12.8 (-21.0 to -4.4) percentage points). The largest changes with both incentive introduction and withdrawal were for complex process indicators such as diabetes foot screening, with smaller changes in simple processes such as blood pressure measurement, intermediate outcomes, and treatment indicators. For all types of indicators, the reduction in quality following incentive withdrawal generally matched or exceeded the gains observed after incentive introduction (for example, for 14 indicators with data for both, median change at three years for incentive introduction was a 1.4 (IQR -0.9 to 4.6) percentage point increase versus a 3.9 (IQR 2.2 to 11.6) decrease for incentive withdrawal).

median change compared with that predicted by pre-

incentivisation trends 6.1 (interguartile range (IQR)

1.9 to 14.6) percentage points) but less consistently

CONCLUSION

Quality and Outcomes Framework incentives consistently improved quality of care at one year beyond that predicted by pre-incentivisation trends, but by three years the impact was inconsistent and not clearly better than trend. Gains from incentivisation seemed to reverse after incentive withdrawal.

STUDY REGISTRATION

Prospero CRD42023467627.

Introduction

"Pay for performance" is commonly deployed in healthcare internationally as a tool to improve quality of care. Although physicians and providers clearly respond to both explicit and implicit financial incentives, the effectiveness of pay for performance in improving quality remains uncertain.¹⁻³ Pay for performance often seems to improve the recording of care processes, but evidence that introducing incentives improves intermediate outcomes (such as whether blood pressure is controlled) is inconsistent, with some studies showing benefit and others no benefit. Many studies also examine only short term (for example, one year) impact meaning that longer term effects are often uncertain.1 ² Quality of care measured using quality indicators often worsens when incentives are withdrawn.⁴⁻⁶ For both introduction and withdrawal of incentives, distinguishing whether

observed changes are due to better recording of care already being given or to changes in the care actually delivered is not straightforward. Variation in evidence of impact at least partly reflects heterogeneity in the design of evaluations, with weaker designs typically finding larger benefits of incentive introduction because they do not properly account for preincentivisation trends.²⁷

The UK Quality and Outcomes Framework (QOF) introduced across the National Health Service (NHS) in April 2004 remains one of the largest ever primary care pay-for-performance programmes in the world (supplementary box A). In its original form, it included more than 150 indicators and accounted for 20-25% of the income of UK primary care practices (which are typically small, physician owned organisations that contract to provide publicly funded NHS care). Over time, several indicators were retired and new indicators introduced, and the proportion of income tied to the QOF reduced. A large number of indicators were removed at the end of March 2014, and at the end of March 2016 NHS Scotland removed all QOF incentivisation. The QOF therefore provides a systemwide natural experiment to examine the impact of introducing and withdrawing financial incentives. The aim of this study was to systematically review studies of the impact of UK QOF incentives on quality of primary care, including quantitative synthesis of the one and three year impact of both incentive introduction and incentive withdrawal.

Methods

Study overview

We registered the protocol in PROSPERO (CRD42023467627) and report the review on the basis of the PRISMA (Preferred Reporting Items for Systematic Reviews and Meta-Analyses) guidelines.⁸ This paper reports findings for the first two registered review questions. (1) What was the impact on measured quality when incentives were introduced? (2) What was the impact on measured quality when incentives were withdrawn? For comparison we also report data for changes in indicators that were not incentivised, before and after introduction of the QOF in April 2004.

Eligibility criteria

Studies on introduction and withdrawal of incentives were eligible if they used repeated cross sectional or cohort designs with consistent measurement of quality of care in defined UK populations before and after incentive introduction or withdrawal. We had two requirements for inclusion in the primary quantitative synthesis. Firstly, the measures examined had to be either the same as or very closely related to the incentivised indicator (for example, the QOF incentivised care for "diabetes" without specifying type, but we included data for the same quality indicators examined in people with type 1 diabetes) or directly related to it (for example, the QOF incentivised advising women of the benefits of long acting reversible contraception where the explicit intent was to increase its use, and we included indicators of the uptake of long acting reversible contraception). Secondly, we required studies to have a minimum of three time points before and three after the intervention, as estimating trends is otherwise not possible.

The narrative synthesis included other studies with consistently measured data before and after incentive introduction/withdrawal but which had too few time points for interrupted time series analysis and/or which did not report (or plot) raw time series data. We excluded studies in which the population studied or quality measurement was inconsistent over time, single time point before-and-after studies without any control, studies of non-QOF financial incentives, conference abstracts, systematic and umbrella reviews, and editorials.

Search strategy and study selection

We searched Medline, Embase, CINAHL, PsycINFO, and Scopus from 1 January 2004 to 3 September 2024, with keywords for "QOF", "primary care", "pay-forperformance", and "United Kingdom" (supplementary box B details search strategies), with additional hand searching of reference lists of included studies. We imported all records into Covidence (Veritas Health Innovation, Melbourne, Australia) for deduplication, followed by title and abstract screening and full text assessment conducted independently by two reviewers. Conflicts in the screening and assessment were resolved through discussion between the two reviewers with arbitration from a third reviewer if required.

Data extraction

We extracted key characteristics from all included studies by using a pre-developed, piloted data extraction sheet. Extracted study characteristics included publication year, region/country, data source, study intervention, study design, time frame, evaluation method, quality indicators evaluated, and relevant study data (including one and three year absolute impact if evaluated or underlying time series data if not).

Where studies reported one and three year impact, we extracted these findings from the paper and used them in analysis. One study of incentive introduction used a multilevel regression model to calculate impact.9 One study of incentive introduction used interrupted time series analysis and reported one and four year impact for one indicator (we included reported four year impact in the quantitative synthesis of three year impact). Two studies of incentive withdrawal used interrupted time series analysis,⁴⁵ one of which was a controlled interrupted time series analysis (change in Scotland where incentives were withdrawn controlled by changes in England where incentives were maintained; we include the reported controlled interrupted time series analysis results).⁵

Where studies did not report one and three year impact, we extracted raw time series data from tables where available (two studies¹⁰¹¹). Where time series data were not reported in tables, we extracted raw data from time series plots by using plotdigitizer software (https://plotdigitizer.com/) (five studies¹²⁻¹⁶). This de novo reanalysis was not explicitly stated in the pre-registered protocol (https://www.crd.york.ac.uk/prospero/display_record.php?ID=CRD42023467627) as we decided on this when it became clear that multiple studies had extractable data for which de novo interrupted time series analysis would estimate impact consistent with that reported by other included studies and therefore improve coverage in the quantitative synthesis.

De novo interrupted time series analysis methods

For each time series, we fitted models to the extracted raw data. Model choices were consistent with the original analyses, in terms of using their pre-specified timing of the intervention and whether any of the pre-intervention data points were omitted from modelling (the QOF had a "preparatory" year in 2003-04 before incentivisation when practices knew the indicators that would be used, so for some indicators performance changed beyond the previous trend in this preparatory year). We fitted interrupted time series analysis models using the itsa command in StataBE v18,¹⁷ with one and three year impact estimated using the lincom command.¹⁸ For time series with more than three pre-intervention time points, we explored whether autocorrelation was present by using the actest command and fitted lag terms to account for autocorrelation where needed.

For indicators with high pre-intervention performance and upward trends, predicted performance (based on previous trend) at three years could exceed 100%. In this situation, we did not estimate three year impact, but pay for performance in these circumstances cannot usually have a major impact because performance is approaching a ceiling anyway.

Risk of bias assessment

One reviewer (LH) assessed risk of bias by using the methodological quality criteria for quantitative nonrandomised studies of the Mixed Methods Assessment Tool (MMAT). A second reviewer (BG) re-scored risk of bias during data extraction but not masked to LH's assessment, with discrepancies resolved by discussion.¹⁹ All of the studies included in the quantitative synthesis fulfilled all criteria for the risk of bias assessment by the MMAT (supplementary table C), meaning that they recruited representative patients/participants, conducted appropriate outcome measurements, reported outcome data completely, and accounted for confounders in data analysis.

Quantitative synthesis

Our starting assumption was that findings should not be meta-analysed, as the diversity of indicators being incentivised meant that we did not believe that a "true" single value of pay-for-performance impact existed that could be estimated.²⁰ Consistent with Cochrane Collaboration recommendations, we therefore chose to report the distribution of observed impact of incentivisation and removal of incentives for all indicators and in selected subgroups by using both quantitative synthesis (median and interquartile range of estimated impact) and graphical visualisation.²⁰ We examined how impact of incentive introduction and withdrawal varied by baseline performance (value of the quality indicator in the year before the intervention) by using graphical visualisation and by estimating Pearson's correlation coefficient. Synthesis was done separately for incentive introduction and incentive withdrawal. One paper also reported change in quality for unincentivised indicators before and after the QOF was introduced in April 2004, and we report these data as well for comparison.²¹

We categorised subgroup analysis by indicator type (simple processes such as blood pressure recording that could be done opportunistically in any consultation, complex processes such as diabetes retinal screening that could not usually be done opportunistically, intermediate outcomes such as blood pressure control, and treatments such as influenza immunisation) and by type of condition (cardiovascular disease, diabetes, and other conditions). We additionally examined impact of incentive introduction and withdrawal for 14 sets of indicators for which data were available for both. We used narrative synthesis for eligible studies in which one and three year impact was neither reported nor re-analysable.

Patient and public involvement

The idea for the review was informed by discussions with public contributors about primary care reform in Scotland, and public contributors supported the writing of the lay summary infographic shared on social media (@acrcedincare) and policy briefing (https://doi.org/10.5281/zenodo.15372055).

Results

The literature search after deduplication yielded 3495 records, and we screened 100 full text records for the two pre-registered research questions reported here. Thirty studies were eligible for analysis. Twenty four studies focused on introduction of incentives (including one also studying impact on non-incentivised indicators), three on withdrawal of incentives, and three on both introduction and withdrawal (supplementary figure A: supplementary tables A and B). Eleven studies were included in the quantitative synthesis. Risk of bias was low for all studies included in the quantitative synthesis (supplementary table C). Seven (35%) studies included in the narrative synthesis were judged to be at risk of bias-one because it failed to satisfy the criteria for participant representativeness as it was conducted in one particular region and six because of incomplete accounting for confounders (specifically in relation to accounting for pre-intervention trends when estimating impact).

Table 1 Change in reported quality at one year and three years				
	1 year		3 years	
		Median (IQR) impact or		Median (IQR) impact or
	No of indicators	change*	No of indicators†	change*
All indicators				
Incentive introduction	83	6.1 (1.9 to 14.6)	72	0.7 (-2.1 to 8.9)
Never incentivised	19	0.0 (-1.1 to 1.4)	19	-1.9 (-5.5 to 0.1)
Incentive withdrawn	31	-10.7 (-17.9 to -3.8)	31	-12.8 (-21.0 to -4.4)
Simple process				
Incentive introduction	38	5.8 (2.2 to 13.8)	28	0.6 (-2.8 to 5.7)
Never incentivised	9	0.3 (-0.9 to 1.5)	9	-4.3 (-6.4 to -0.3)
Incentive withdrawn	7	-16.0 (-24.2 to -11.5)	7	-18.0 (-24.6 to -10.8)
Complex process				
Incentive introduction	16	24.3 (13.8 to 31.0)	15	10.1 (-2.0 to 18.3)
Never incentivised	0	-	0	-
Incentive withdrawn	7	-36.6 (-54.6 to -22.4)	7	-46.1 (-59.8 to -31.5)
Intermediate outcome				
Incentive introduction	14	2.6 (0.4 to 5.3)	14	-0.7 (-4.9 to 0.4)
Never incentivised	0	-	0	-
Incentive withdrawn	11	-8.0 (-10.6 to -4.7)	11	-12.7 (-15.2 to -7.6)
Treatment and immunisation				
Incentive introduction	15	3.8 (0.6 to 7.6)	15	3.2 (0.4 to 8.6)
Never incentivised	10	-0.1 (-2.4 to 1.0)	10	-0.2 (-4.4 to 0.2)
Incentive withdrawn	6	-3.2 (-3.7 to -1.4)	6	-2.8 (-3.4 to -1.6)
Cardiovascular conditions				
Incentive introduction	15	5.8 (2.6 to 11.4)	14	1.2 (-0.1 to 2.6)
Never incentivised	7	-0.01 (-2.2 to 0.8)	7	-5.0 (-7.4 to -3.2)
Incentive withdrawn	11	-10.5 (-14.2 to -6.0)	11	-13.7 (-17.4 to -7.0)
Diabetes				
Incentive introduction	56	6.0 (1.8 to 11.1)	46	-0.7 (-5.9 to 7.4)
Never incentivised	1	-0.3 (-0.3 to -0.3)	1	-0.3 (-0.3 to -0.3)
Incentive withdrawn	8	-4.8 (-8.7 to -3.0)	8	-7.7 (-13.2 to -3.2)
Other conditions				
Incentive introduction	12	13.6 (2.7 to 23.0)	12	8.7 (0.7 to 14.5)
Never incentivised	11	0.2 (-0.5 to 1.6)	11	-0.1 (-3.8 to -0.4)
Incentive withdrawn	12	-22.2 (-32.4 to -9.6)	12	-24.6 (-41.7 to -9.6)
Paired indicators				
Incentive introduction	16	3.8 (1.2 to 6.4)	16	1.4 (-0.9 to 4.6)
Incentive withdrawal	15	-3.9 (-9.2 to -3.2)	15	-3.9 (-11.6 to -2.2)

IQR=interguartile range

*Positive value indicates higher observed quality than predicted by previous trend; negative value indicates lower observed quality than predicted by previous trend.

†Number is less than for one year because not all studies reported three year impact and/or de novo interrupted time series analysis could not estimate three year impact because predicted performance exceeded 100%.

Quantitative synthesis of impact on measured quality

One and three year absolute percentage point change in indicators was reported or could be estimated for 83 indicators for which incentives were introduced, for 31 indicators for which incentives were withdrawn, and for 19 unincentivised indicators measured before and after introduction of the QOF in 2004 (supplementary tables D and E; supplementary figure B).

For all indicators, we found evidence that introduction of the incentive was usually associated with higher reported quality at one year compared with expected levels based on pre-incentivisation trends (median change 6.1 (interquartile range (IQR) 1.9 to 14.6) percentage points; 74/83 indicators improved) but not at three years (median change 0.7 (-2.1 to 8.9); 40/72 indicators improved) (table 1; fig 1; supplementary figures B-D). For incentive removal, reported quality reduced compared with expected at both one year (median change -10.7 (IQR -17.9 to

-3.8)) and three years (median change -12.8 (-21.0 to -4.4). For indicators that were never incentivised, we found no consistent evidence of change from expected at one year (median change 0.0 (IQR -1.1 to 1.4)), but quality was modestly worse than expected at three years (-1.9 (-5.5 to 0.1)).

For incentive introduction, impact fell with increasing baseline performance in the year before introduction at both one year (Pearson r=-0.772; P<0.001) and three years (Pearson r=-0.627; P<0.001) (fig 2). For incentive withdrawal, no correlation existed between withdrawal and impact at either one year (Pearson r=0.164; P=0.39) or three years (Pearson r=0.132; P=0.49).

For simple process indicators (table 1; fig 3), incentive introduction was associated with improved recorded quality at one year compared with expected (median change 5.8 (IQR 2.2 to 13.8) percentage points) but not at three years (median change 0.6 (-2.8 to 5.7) percentage points). Incentive withdrawal



Fig 1 | Change in recorded quality at one and three years after introduction and removal of incentives and for never incentivised indicators. Indicators ordered in descending order of one year impact/change, paired with three year impact/change on same vertical. Supplementary figures C and D show these data with indicator names. IQR=interquartile range

was associated with larger reductions than expected at both one year (median change -16.0 (IQR -24.2to -11.5)) and three years (median change -18.0(-24.6 to -10.8)). By contrast, unincentivised simple process indicators at the time of QOF introduction in 2004 showed little change at one year but evidence of decline from expected at three years (median change -4.3 (IQR -6.4 to -0.3)).

Improvements from incentivisation of complex processes such as diabetic retinal screening were larger than those of simple processes at one year (median change 24.3 (IQR 13.8 to 31.0) percentage points)



Fig 2 | Impact of incentive introduction and withdrawal in relation to baseline performance in year before intervention. Pearson's correlation coefficient calculated ignoring outlier points (marked with triangle), which are studies of introduction or withdrawal of incentives for long acting reversible contraception (LARC) with measures expressed as percentage of all women prescribed LARC



Fig 3 | Impact of incentive introduction and incentive withdrawal at one and three years, and change in never incentivised indicators. Eleven indicators evaluable at one year are not evaluated at three years because predicted quality at three years exceeds 100%; both plots ordered in ascending order of impact at either one year or three years. Statistical significance is based on 95% confidence interval for estimate of impact not including zero, so equivalent to P<0.05

and three years (median change 10.1 (-2.0 to 18.3) percentage points). However, withdrawal of incentives for seven complex process indicators was associated with very large reductions in reported quality at three years (median change -46.1 (range -59.8 to -31.5)). Observed benefit of incentive introduction for intermediate outcomes such as blood pressure control was small and usually not statistically significant, but reported quality of intermediate outcomes worsened compared with expected when incentives were withdrawn (at three years, median change -12.7 (range -15.2 to -7.6)).

For treatment indicators, median change for incentive introduction was a small median improvement at one year (median change 3.8 (IQR 0.6 to 7.6) percentage points) and at three years (3.2 (0.4 to 8.6)), although all of the largest improvements were for five influenza immunisation indicators. For withdrawal of incentives for treatment indicators, median change was -3.2 (IQR -3.7 to -1.4) percentage points at one year and -2.8 (-3.4 to -1.6) at three years. For 10 never incentivised treatment indicators, we found little evidence of change (at three years, median change -0.2 (IQR -4.4 to 0.2)).

Patterns of change were similar to the main findings for cardiovascular and diabetes indicators but somewhat larger in magnitude for the 33 indicators relating to various other conditions (at one year, median change for incentive introduction 13.6 (IQR 2.7 to 23.0) percentage points; for incentive withdrawal -22.2(-32.4 to -9.6) (table 1; supplementary figure E).

For 14 groups of indicators, data on impact was available for both incentive introduction and incentive withdrawal (table 1; fig 4; supplementary table F). We found variation in observed absolute changes but consistent evidence of improved reported quality compared with expected at one year with incentivisation (median change 3.8 (IQR 1.2 to 6.4) percentage points) and of reduced reported quality with incentive removal (median change -3.9 (-9.2 to -3.2)). Benefits of incentivisation compared with expected were smaller at three years (median 1.4 (IQR -0.9 to 4.6)), whereas median change on incentive removal was similar (-3.9 (-11.6 to -2.2)).

Other studies of impact

A study using case note review with two time points before and two after found that in the year after



Fig 4 | Impact of incentive introduction and withdrawal at one and three years for same/similar indicators. A: incentives introduced for blood pressure (BP) $\leq 145/85$ mm Hg; incentive withdrawal evaluated at two levels ($\leq 150/90$ mm Hg and $\leq 140/80$ mm Hg). B: incentives introduced for coronary heart disease (CHD) β blocker treatment and for CHD with left ventricular systolic dysfunction angiotensin converting enzyme inhibitor/angiotensin receptor blocker treatment; incentives withdrawn for CHD antithrombotic treatment. C: incentive introduction evaluated for glycated haemoglobin (HbA_{1c}) $\leq 7.4\%$; incentive withdrawal evaluated for HbA_{1c} ≤ 59 mmol/mol (7.5%). D: incentive introduction evaluated for HbA_{1c} $\leq 10\%$; incentive withdrawal evaluated for HbA_{1c} ≤ 75 mmol/mol (9%). Statistical significance is based on 95% confidence interval for estimate of impact not including zero, so equivalent to P<0.05. COPD=chronic obstructive pulmonary disease; LARC=long acting reversible contraception; TIA=transient ischaemic attack

implementation the QOF was not associated with any improvement in recorded quality for coronary heart disease, but a statistically significant increase occurred in recorded quality for asthma and diabetes. For all three conditions, performance did not improve further between one and three years, which the authors interpreted as being a result of ceiling effects. Continuity of care assessed by surveying patients statistically significantly worsened after QOF introduction, although absolute changes were small (supplementary table G).⁹

Other studies that could not be quantitatively synthesised found that introduction of incentives was associated with increased consultation by patients with incentivised conditions but not associated with any change in consultations for all patients in the practice.²²⁻²⁴ Processes of care were usually estimated to improve after incentivisation, but not consistently.¹⁶ ²⁵⁻³⁰ Some evidence also suggested that incentivising quality in one group of people was

associated with improvements in the same activity in groups in which incentives did not apply ("halo effects") for risk factor recording,²⁵ smoking cessation advice,²⁸ and mean daily number of cigarettes smoked.³¹ Evidence that QOF incentives to record risk factors or control intermediate outcomes changed prescribing was mixed. For people with diabetes, blood pressure control and cholesterol control improved, but glycated haemoglobin control got worse,³² although other studies found evidence of earlier introduction of hypoglycaemic drugs.³³ No evidence supported a change in prescribing in hypertension or in initiation of cardioprotective drugs in people with serious mental illness despite improved risk factor recording.^{29 34 35} Incentivising the use of depression severity measures at diagnosis of depression was associated with a reduction in first episodes of depression treated with antidepressants.³⁶ One study of withdrawal of incentives for recording cardiovascular risk factors in people with serious mental illness found large absolute

reductions in recording,³⁴ whereas another found little impact on simple processes of care in multiple conditions (although the process indicators examined remained incentivised by other indicators).³⁷

In relation to broader outcomes, trend adjusted QOF related ambulatory care sensitive admissions fell by 2.7% in the first year of the QOF compared with unrelated admissions and by 2.8% compared with non-ambulatory care sensitive admissions, increasing to 7.7% and 8.7% reductions respectively by year three, although the QOF was not the only relevant policy change in the period.³⁸ However, UK QOF related mortality did not decline faster than expected compared with changes in a matched set of other countries.³⁹

Discussion

The introduction of QOF financial incentives was associated with one year improvements in recorded care processes compared with levels expected on the basis of pre-incentivisation trends. However, by three years performance on simple process indicators was not consistently better than expected, and improvements in complex processes were somewhat smaller than those observed at one year. Improvements in intermediate outcomes over preincentivisation trends were relatively small at one year and minimal at three years. Although improvements beyond trend in incentivised treatment were small, they persisted at three years. We found some evidence that improvements on introduction of incentives were potentially constrained by "ceiling effects" (whereby quality is close to a maximum before incentivisation), as the observed impact was smaller when baseline performance was higher. Withdrawal of incentives led to reductions in recorded quality of care, often matching or exceeding any gains associated with incentive introduction. We found no strong evidence of impact on unincentivised care at one year, but by three years some evidence suggested small declines in quality of unincentivised care compared with preincentivisation trends (potential "crowding out" of unincentivised care). One national study provided strong evidence of reductions in related ambulatory care sensitive admissions after implementation of the QOF (although causality is uncertain), but no change in QOF related mortality was seen.

Strengths and limitations of study

Strengths of the study include systematic evaluation of the short and medium term impact of incentivising and de-incentivising multiple indicators as part of a national pay-for-performance programme with large financial incentives. The study has several limitations. Firstly, interpretation of incentive withdrawal is reasonably simple because recorded quality was stable before incentives were withdrawn,^{4 5} so observed changes typically represent actual reductions from a previously steady state. By contrast, interpretation of incentive introduction is usually in the context of improving recorded quality, driven by other quality improvement work (such as national and local guideline implementation and national service frameworks) to which financial incentives were added. Compared with incentive withdrawal, therefore, a stronger assumption is that quality would have continued to improve at the same rate without financial incentives. Whether existing improvement work would have delivered the same level of improvement without incentives is not known, and at least some indicators were subject to ceiling effects that limited the scope for continued improvement. This is most obvious in evaluation of three year impact, in which predicted quality sometimes exceeded 100% (in this situation, we chose not to estimate three year impact of incentive introduction as the counterfactual is implausible). Ceiling effects potentially mean that the impact of incentive introduction is being underestimated, although patterns are consistent even for indicators for which recorded quality is some distance from 100% (supplementary figure B).

Secondly, evaluation of intermediate outcome indicators is problematic because under the QOF. failure to record an intermediate outcome is assumed to mean that the outcome target is not achieved. This means that changes in the process indicator may lead to changes in the intermediate outcome indicator even if actual (but unrecorded) intermediate outcomes are the same. Caution is therefore needed when interpreting impact for intermediate outcome indicators. Similar considerations apply to some complex processes that were not routinely recorded before incentivisation (such as pre-conceptual advice in women with epilepsy). As this care was not routinely recorded historically, no evaluations of incentive introduction are available, but incentive withdrawal shows very large reductions in reported quality that may just represent a return to not routinely recording this activity. Caution is therefore needed in directly comparing the impact of introduction and withdrawal because the indicators being evaluated are different. However, our paired indicator analysis is consistent with the overall findings.

Thirdly, although related ambulatory care sensitive admissions fell after implementation of the QOF, this was not the only improvement programme in the UK that covered these, so interpretation of this finding should be cautious. These considerations highlight that "recorded quality" can have a variable relation with actual delivery of care. Finally, the focus of this analysis has been incentivised indicators, which risks ignoring harms of pay-for-performance programmes. This review found some evidence of "crowding out" of unincentivised care for 19 indicators reported by one study,²¹ and that patients' perceptions of continuity worsened.⁹ Consistent with this, multiple qualitative studies have found that many general practice staff perceived that the focus on QOF indicators led to less attention being paid to both unincentivised aspects of care and to patients' concerns, with perceived risk of reductions in personal continuity.40 Given QOF implementation in a resource constrained system,

tunnel vision and crowding out are likely to have occurred more widely and are likely in any substantial pay-for-performance programme.⁴¹

Comparison with other studies

The findings are consistent with previous international narrative reviews which concluded that introduction of pay for performance has inconsistent benefit, with improvements largely observed in the short term for processes of care and larger impact when baseline performance was lower.¹² Incentive withdrawal is less well studied outside of the QOF, but two randomised trials of hypertension care in the US and chlamydia screening in Australia also found that improvements following incentivisation were not sustained when incentives were withdrawn,⁶⁴² which was also observed in a UK non-OOF programme of incentivisation of screening and brief intervention for alcohol misuse.⁴³ The QOF used financial incentives that were larger than in most other countries,² implemented indicators that usually had wide professional acceptance (although baseline performance was therefore often already high), aligned incentives to reputational incentives through public reporting,⁴⁴ and provided implementation and educational support.^{45 46} Practices seemed to respond with several organisational changes,^{40 47} ⁴⁹ but despite initial gains in recorded quality this study is not consistent with the QOF driving sustained quality improvement over and above existing trends. Notably, performance on many indicators was already improving before incentivisation (because of other quality improvement initiatives such as professional education and national clinical guidelines, with consequent increasing implementation of registers, recall systems, and structured review of people with chronic conditions); and, for some indicators with high baseline performance, impact was likely subject to ceiling effects. Any improvements seen with QOF incentivisation were also not sustained in the sense that withdrawal of incentives was associated with reductions in recorded quality. This may have been because the external financial incentives of the QOF reduced internal motivation to maintain quality; it suggests that reorganisation to deliver QOF reporting requirements may not have been optimised to deliver sustainable quality improvement and may have overshadowed other improvement activity.⁴⁰

Some of the gains in recorded quality with incentive introduction (and losses after withdrawal) probably simply reflect changes in recording and in some cases indicator gaming (an example of which is that blood pressure recordings which contributed to QOF payment were systematically lower, and at the payment target, compared with blood pressure recordings at other times of the year which did not contribute to payment).⁵⁰

Implications

The findings of this study emphasise that pay for performance is no magic bullet to improve quality. Pay-for-performance programme designers face the challenge of knowing whether their programme delivers meaningful improvement and whether incentives are driving organisational change that will sustain improvement when incentives are withdrawn. Deployment of pay for performance therefore needs to evaluate the impact of incentive introduction and withdrawal, ideally by independent measurement rather than relying on data reported for payment. More broadly, whether or not the QOF "worked" is a complicated question, because it was never purely a quality improvement initiative. The original bargain between government and general practitioners was that general practices would deliver higher quality of care and in return would receive increased income (which both sides agreed was needed to tackle low morale and problems of recruitment and retention of general practitioners).^{45 46} Both sides of this bargain were delivered, but over time government came to believe that it was locked into paying for steady state (rather than improving) quality, whereas general practitioners saw steady declines in practice income after adjustment for inflation, with falling morale, increasing intention to leave practice, and resurgent problems of recruitment and retention.⁵¹

More broadly, in the 20 years since the OOF was implemented, the challenges posed to health systems by population ageing and the rising prevalence of multiple long term conditions and frailty have been increasingly recognised. The OOF remains one of the largest ever attempts to improve healthcare quality by using pay for performance, using the approach of deploying financial incentives to improve narrow, disease focused indicators. Given that the OOF did not seem to drive sustained change in these narrow indicators after incentives were withdrawn, an unmodified QOF approach is unlikely to be useful for increasing the delivery of very complex behaviours such as care coordination and management of complex problems such as polypharmacy.^{52 53} However, pay for performance or other kinds of financial incentive might have a role to play if deployment aligns them with other improvement mechanisms such as outreach professional education,54 audit and feedback,55 or collaborative improvement models,⁵⁶ which may have more persistent effects on quality of care.42 Notably, more recent QOF incentives in England include payment for participation in national quality improvement programmes rather than delivery of narrow, disease focused indicators.⁵⁷ Such approaches may be better at engaging and sustaining professionals' internal motivation to improve care than narrow payfor-performance programmes,⁵⁸ although this requires further evaluation.

Conclusion

The UK primary care pay-for-performance programme initially improved quality of care as measured by the indicators examined in this study beyond that predicted by pre-incentivisation trends, but improvements associated with incentivisation were not persistent and seemed to reverse after withdrawal of incentives. This paper reports findings related to the first two research questions in the registered protocol (PROSPERO CRD42023467627 https://www.crd.york.ac.uk/prospero/display_record. php?ID=CRD42023467627).

Contributors: BG and SWM conceived the study. LH led literature searching, title/abstract and full text screening (to which all authors contributed), and data extraction. BG did the interrupted time series (re)analysis. All authors contributed to interpretation of the findings. BG and LH initially drafted the paper, with all authors reviewing drafts, contributing to the writing, and approving the final version. BG is the guarantor. The corresponding author attests that all listed authors meet authorship criteria and that no others meeting the criteria have been omitted.

Funding: The study is funded by Economic and Social Research Council (grant reference: ES/T014164/1), with additional funding from the Legal and General Group (research grant to establish the independent Advanced Care Research Centre at University of Edinburgh). The research team is independent of the funders, which had no role in the design and conduct of the study, the analysis and interpretation of findings, the writing of the paper, or the decision to submit it for publication. The views expressed are those of the authors and not necessarily those of the funders.

Competing interests: All authors have completed the ICMJE uniform disclosure form at www.icmje.org/disclosure-of-interest/ and declare: support from the Economic and Social Research Council and Legal and General Group; no financial relationships with any organisations that might have an interest in the submitted work in the previous three years; no other relationships or activities that could appear to have influenced the submitted work.

Ethical approval: The study involved only reanalysis of published, non-disclosive data, and ethical review was not required.

Data sharing: Extracted raw data used in interrupted time series analysis (ITSA) modelling and Stata code used in ITSA analysis are available from https://doi.org/10.5281/zenodo.14951135.

Transparency: The lead author (the manuscript's guarantor) affirms that this manuscript is an honest, accurate, and transparent account of the study being reported; that no important aspects of the study have been omitted; and that any discrepancies from the study as planned (and, if relevant, registered) have been explained.

Dissemination to patient and public communities: Public contributors supported the writing of the lay summary shared on social media (@acrcedincare) and policy briefing (https://doi. org/10.5281/zenodo.15372055).

Provenance and peer review: Not commissioned; externally peer reviewed

This is an Open Access article distributed in accordance with the terms of the Creative Commons Attribution (CC BY 4.0) license, which permits others to distribute, remix, adapt and build upon this work, for commercial use, provided the original work is properly cited. See: http://creativecommons.org/licenses/by/4.0/.

- 1 Eijkenaar F, Emmert M, Scheppach M, Schöffski O. Effects of pay for performance in health care: a systematic review of systematic reviews. *Health Policy* 2013;110:115-30. doi:10.1016/j. healthpol.2013.01.008
- 2 Mendelson A, Kondo K, Damberg C, et al. The effects of pay-forperformance programs on health, health care use, and processes of care: A systematic review. *Ann Intern Med* 2017;166:341-53. doi:10.7326/M16-1881
- 3 Wagenschieber E, Blunck D. Impact of reimbursement systems on patient care - a systematic review of systematic reviews. *Health Econ Rev* 2024;14:22. doi:10.1186/s13561-024-00487-6
- 4 Minchin M, Roland M, Richardson J, Rowark S, Guthrie B. Quality of Care in the United Kingdom after Removal of Financial Incentives. N Engl J Med 2018;379:948-57. doi:10.1056/ NEJMsa1801495
- 5 Morales DR, Minchin M, Kontopantelis E, Roland M, Sutton M, Guthrie B. Estimated impact from the withdrawal of primary care financial incentives on selected indicators of quality of care in Scotland: controlled interrupted time series analysis. *BMJ* 2023;380:e072098. doi:10.1136/bmj-2022-072098
- 6 Petersen LA, Simpson K, Pietz K, et al. Effects of individual physicianlevel and practice-level financial incentives on hypertension care: a randomized trial. *JAMA* 2013;310:1042-50. doi:10.1001/ jama.2013.276303
- 7 Zaresani A, Scott A. Is the evidence on the effectiveness of pay for performance schemes in healthcare changing? Evidence from a meta-regression analysis. *BMC Health Serv Res* 2021;21:175. doi:10.1186/s12913-021-06118-8

- 8 Page MJ, McKenzie JE, Bossuyt PM, et al. The PRISMA 2020 statement: an updated guideline for reporting systematic reviews. BMJ 2021;372:n71. doi:10.1136/bmj.n71
- 9 Campbell SM, Reeves D, Kontopantelis E, Sibbald B, Roland M. Effects of pay for performance on the quality of primary care in England. N Engl J Med 2009;361:368-78. doi:10.1056/ NEJMsa0807651
- 10 Kontopantelis E, Reeves D, Valderas JM, Campbell S, Doran T. Recorded quality of primary care for patients with diabetes in England before and after the introduction of a financial incentive scheme: a longitudinal observational study. *BMJ Qual Saf* 2013;22:53-64. doi:10.1136/bmjqs-2012-001033
- 11 Calvert M, Shankar A, McManus RJ, Lester H, Freemantle N. Effect of the quality and outcomes framework on diabetes care in the United Kingdom: retrospective cohort study. *BMJ* 2009;338:b1870. doi:10.1136/bmj.b1870
- 12 Arrowsmith ME, Majeed A, Lee JT, Saxena S. Impact of pay for performance on prescribing of long-acting reversible contraception in primary care: an interrupted time series study. *PLoS One* 2014;9:e92205. doi:10.1371/journal.pone.0092205
- 13 Simpson CR, Hannaford PC, Ritchie LD, Sheikh A, Williams D. Impact of the pay-for-performance contract and the management of hypertension in Scottish primary care: a 6-year population-based repeated cross-sectional study. Br J Gen Pract 2011;61:e443-51. doi:10.3399/bjgp11X583407
- 14 Pasvol TJ, Macgregor EA, Rait G, Horsfall L. Time trends in contraceptive prescribing in UK primary care 2000-2018: a repeated cross-sectional study. *BMJ Sex Reprod Health* 2022;48:193-8. doi:10.1136/bmjsrh-2021-201260
- 15 Taggar JS, Coleman T, Lewis S, Szatkowski L. The impact of the Quality and Outcomes Framework (QOF) on the recording of smoking targets in primary care medical records: cross-sectional analyses from The Health Improvement Network (THIN) database. BMC Public Health 2012;12:329. doi:10.1186/1471-2458-12-329
- 16 Matias MA, Jacobs R, Aragón MJ, et al. Assessing the uptake of incentivised physical health checks for people with serious mental illness: a cohort study in primary care. Br J Gen Pract 2024;74:e449-55. doi:10.3399/BJGP.2023.0532
- 17 Linden A. Conducting interrupted time-series analysis for singleand multiple-group comparisons. *Stata J* 2015;15:480-500. doi:10.1177/1536867X1501500208
- 18 Linden A. A comprehensive set of postestimation measures to enrich interrupted time-series analysis. *Stata J* 2017;17:73-88. doi:10.1177/1536867X1701700105
- 19 Hong Q, Pluye P, Fàbregues S, et al. Mixed Methods Appraisal Tool (MMAT), version 2018. 2018. http:// mixedmethodsappraisaltoolpublic.pbworks.com/w/file/ fetch/127916259/MMAT_2018_criteria-manual_2018-08-01_ ENG.pdf
- 20 McKenzie J, Brennan S. Chapter 12: Synthesizing and presenting findings using other methods. Cochrane Handbook for Systematic Reviews of Interventions Version 65. 2024. https://training.cochrane. org/handbook/current/chapter-12
- 21 Doran T, Kontopantelis E, Valderas JM, et al. Effect of financial incentives on incentivised and non-incentivised clinical activities: longitudinal analysis of data from the UK Quality and Outcomes Framework. *BMJ* 2011;342:d3590. doi:10.1136/bmj.d3590
- 22 Kontopantelis E, Olier I, Planner C, et al. Primary care consultation rates among people with and without severe mental illness: a UK cohort study using the Clinical Practice Research Datalink. *BMJ Open* 2015;5:e008650. doi:10.1136/bmjopen-2015-008650
- 23 James GD, Donaldson GC, Wedzicha JA, Nazareth I. Trends in management and outcomes of COPD patients in primary care, 2000-2009: a retrospective cohort study. NPJ Prim Care Respir Med 2014;24:14015. doi:10.1038/npjpcrm.2014.15
- 24 Gilbert C, Allgar V, Doran T. Workload impact of the Quality and Outcomes Framework for patients with diabetes: an interrupted time series in general practice. *Br J Gen Pract* 2019;69:e570-7. doi:10.3399/bjgp19X704645
- 25 Sutton M, Elder R, Guthrie B, Watt G. Record rewards: the effects of targeted quality incentives on the recording of risk factors by primary care providers. *Health Econ* 2010;19:1-13. doi:10.1002/hec.1440
- 26 Hardoon SL, Khadjesari Z, Nazareth I, Hamilton FL, Petersen I. Monitoring of alcohol consumption in primary care among adults with bipolar disorder: A cross-sectional and retrospective cohort study. *J Affect Disord* 2016;198:83-7. doi:10.1016/j. jad.2016.03.019
- 27 McLintock K, Russell AM, Alderson SL, et al. The effects of financial incentives for case finding for depression in patients with diabetes and coronary heart disease: interrupted time series analysis. *BMJ Open* 2014;4:e005178. doi:10.1136/ bmjopen-2014-005178
- 28 Szatkowski L, Aveyard P. Provision of smoking cessation support in UK primary care: impact of the 2012 QOF revision. Br J Gen Pract 2016;66:e10-5. doi:10.3399/bjgp15X688117



- 29 Serumaga B, Ross-Degnan D, Avery AJ, et al. Effect of pay for performance on the management and outcomes of hypertension in the United Kingdom: interrupted time series study. *BMJ* 2011;342:d108. doi:10.1136/bmj.d108
- 30 Khadjesari Z, Hardoon SL, Petersen I, Hamilton FL, Nazareth I. Impact of Financial Incentives on Alcohol Consumption Recording in Primary Health Care Among Adults with Schizophrenia and Other Psychoses: A Cross-Sectional and Retrospective Cohort Study. Alcohol Alcohol 2017;52:197-205.
- 31 Fichera E, Gray E, Sutton M. How do individuals' health behaviours respond to an increase in the supply of health care? Evidence from a natural experiment. *Soc Sci Med* 2016;159:170-9. doi:10.1016/j. socscimed.2016.05.005
- 32 Millett C, Bottle A, Ng A, et al. Pay for perfomance and the quality of diabetes management in individuals with and without co-morbid medical conditions. *J R Soc Med* 2009;102:369-77. doi:10.1258/ jrsm.2009.090171
- Gallagher N, Cardwell C, Hughes C, O'Reilly D. Increase in the pharmacological management of Type 2 diabetes with pay-forperformance in primary care in the UK. *Diabet Med* 2015;32:62-8. doi:10.1111/dme.12575
- 34 Sheppard JP, Stevens S, Stevens RJ, et al. Association of guideline and policy changes with incidence of lifestyle advice and treatment for uncomplicated mild hypertension in primary care: a longitudinal cohort study in the Clinical Practice Research Datalink. *BMJ Open* 2018;8:e021827. doi:10.1136/ bmjopen-2018-021827
- 35 Wilson CL, Rhodes KM, Payne RA. Financial incentives improve recognition but not treatment of cardiovascular risk factors in severe mental illness. *PLoS One* 2017;12:e0179392. doi:10.1371/journal. pone.0179392
- 36 Kendrick T, Stuart B, Newell C, Geraghty AWA, Moore M. Did NICE guidelines and the Quality Outcomes Framework change GP antidepressant prescribing in England? Observational study with time trend analyses 2003-2013. J Affect Disord 2015;186:171-7. doi:10.1016/j.jad.2015.06.052
- 37 Kontopantelis E, Springate D, Reeves D, Ashcroft DM, Valderas JM, Doran T. Withdrawing performance indicators: retrospective analysis of general practice performance under UK Quality and Outcomes Framework. *BMJ* 2014;348:g330. doi:10.1136/bmj.g330
- 38 Harrison MJ, Dusheiko M, Sutton M, Gravelle H, Doran T, Roland M. Effect of a national primary care pay for performance scheme on emergency hospital admissions for ambulatory care sensitive conditions: controlled longitudinal study. *BMJ* 2014;349:g6423. doi:10.1136/bmj.g6423
- 39 Ryan AM, Krinsky S, Kontopantelis E, Doran T. Long-term evidence for the effect of pay-for-performance in primary care on mortality in the UK: a population study. *Lancet* 2016;388:268-74. doi:10.1016/ S0140-6736(16)00276-2
- 40 Khan N, Rudoler D, McDiarmid M, Peckham S. A pay for performance scheme in primary care: Meta-synthesis of qualitative studies on the provider experiences of the quality and outcomes framework in the UK. BMC Fam Pract 2020;21:142. doi:10.1186/s12875-020-01208-8
- 41 Mannion R, Braithwaite J. Unintended consequences of performance measurement in healthcare: 20 salutary lessons from the English National Health Service. *Intern Med J* 2012;42:569-74. doi:10.1111/j.1445-5994.2012.02766.x
- 42 Hocking JS, Wood A, Temple-Smith M, et al. The impact of removing financial incentives and/or audit and feedback on chlamydia testing in general practice: A cluster randomised controlled trial (ACCEPtable). *PLoS Med* 2022;19:e1003858. doi:10.1371/journal. pmed.1003858

- 43 O'Donnell A, Angus C, Hanratty B, Hamilton FL, Petersen I, Kaner E. Impact of the introduction and withdrawal of financial incentives on the delivery of alcohol screening and brief advice in English primary health care: an interrupted time-series analysis. Addiction 2020;115:49-60. doi:10.1111/add.14778
- 44 Allen T, Whittaker W, Kontopantelis E, Sutton M. Influence of financial and reputational incentives on primary care performance: a longitudinal study. *Br J Gen Pract* 2018;68:e811-8. doi:10.3399/ bjgp18X699797
- 45 Roland M, Campbell S. Successes and failures of pay for performance in the United Kingdom. N Engl J Med 2014;370:1944-9. doi:10.1056/NEJMhpr1316051
- 46 Guthrie B, Tang J. What did we learn from 12 years of QOF? 2016. https://www.sspc.ac.uk/media/Media_547662_smxx.pdf
- 47 Huby G, Guthrie B, Grant S, et al. Whither British general practice after the 2004 GMS contract? Stories and realities of change in four UK general practices. *J Health Organ Manag* 2008;22:63-78. doi:10.1108/14777260810862416
- 48 Grant S, Huby G, Watkins F, et al. The impact of pay-for-performance on professional boundaries in UK general practice: an ethnographic study. *Sociol Health Illn* 2009;31:229-45. doi:10.1111/j.1467-9566.2008.01129.x
- 49 Checkland K, Harrison S, McDonald R, Grant S, Campbell S, Guthrie B. Biomedicine, holism and general medical practice: responses to the 2004 General Practitioner contract. *Sociol Health Illn* 2008;30:788-803. doi:10.1111/j.1467-9566.2008.01081.x
- 50 Wilding A, Munford L, Guthrie B, Kontopantelis E, Sutton M. Family doctor responses to changes in target stringency under financial incentives. J Health Econ 2022;85:102651. doi:10.1016/j. jhealeco.2022.102651
- 51 Walker B, Moss C, Gibson J, Sutton M, Spooner S, Checkland K. Tenth National GP Worklife Survey. Policy Research Unit in Commissioning and the Healthcare System, 2019.
- 52 Yordanov D, Oxholm AS, Prætorius T, Kristensen SR. Financial incentives for integrated care: A scoping review and lessons for evidence-based design. *Health Policy* 2024;141:104995. doi:10.1016/j.healthpol.2024.104995
- 53 Lester H, Campbell S. Developing Quality and Outcomes Framework (QOF) indicators and the concept of 'QOFability'. *Qual Prim Care* 2010;18:103-9.
- 54 O'Brien MA, Rogers S, Jamtvedt G, et al. Educational outreach visits: effects on professional practice and health care outcomes. *Cochrane Database Syst Rev* 2007;2007:CD000409. doi:10.1002/14651858.CD000409.pub2
- 55 Ivers N, Jamtvedt G, Flottorp S, et al. Audit and feedback: effects on professional practice and healthcare outcomes. *Cochrane Database Syst Rev* 2012;2012:CD000259. doi:10.1002/14651858. CD000259.pub3
- 56 Kovacevic L, Naik R, Lugo-Palacios DG, Ashrafian H, Mossialos E, Darzi A. The impact of collaborative organisational models and general practice size on patient safety and quality of care in the English National Health Service: A systematic review. *Health Policy* 2023;138:104940. doi:10.1016/j.healthpol.2023.104940
- 57 Bramwell D, Hotham S, Peckham S, Checkland K, Forbes LJL. Evaluation of the introduction of QOF quality improvement modules in English general practice: early findings from a rapid, qualitative exploration of implementation. *BMJ Open Qual* 2022;11:e001960. doi:10.1136/bmjoq-2022-001960
- 58 Dreischulte T, Donnan P, Grant A, Hapca A, McCowan C, Guthrie B. Safer Prescribing--A Trial of Education, Informatics, and Financial Incentives. N Engl J Med 2016;374:1053-64. doi:10.1056/ NEJMsa1508955

Web appendix: Supplementary materials